Author's response to reviews

Title: Evaluation of a new arterial pressure-based cardiac output device requiring no external calibration

Authors:

Christopher Prasser (christopher.prasser@klinik.uni-regensburg.de)
Sylvia Bele (sylvia.bele@klinik.uni-regensburg.de)
Cornelius Keyl (cornelius.keyl@herzzentrum.de)
Stefan Schweiger (stefan.schweiger@klinik.uni-regensburg.de)
Benedikt Trabold (benedikt.trabold@klinik.uni-regensburg.de)
Matthias Amann (matthias.amann@klinik.uni-regensburg.de)
Julia Welnhofer (julia.welnhofer@web.de)
Christoph Wiesenack (christoph.wiesenak@klinik.uni-regensburg.de)

Version: 3 Date: 8 November 2007

Author's response to reviews:

Dear Editorial Team:

08/11/2007

We have now revised our manuscript (MS: 8987606021398905) entitled “Evaluation of a new arterial pressure-based cardiac output device requiring no external calibration” to reflect Chrissie Kouremenou’s suggestions and your formatting requirements.

The authors declare that the manuscript, or part of it, neither has been published nor is currently under consideration for publication by any other journal. The authors state that they agree with and are responsible for the data presented. The undersigned author certifies that any actual or potential conflicts of interest, including financial relationships or sources of funding, have been clearly disclosed.

Yours sincerely,

Christoph Wiesenack, MD